Author's response to reviews

Title: An examination of knowledge, attitudes and practices related to lead exposure in South Western Nigeria

Authors:

Eugenious O Adebamowo (e_adebamowo@yahoo.com)
Oluwole A Agbede (oluwoleagbede@yahoo.com)
Mynepalli K Sridhar (mkcsridhar@yahoo.com)
Clement A Adebamowo (cadebamo@yahoo.com)

Version: 8  Date: 4 March 2006

Author's response to reviews:

We have gone through the manuscript formatting checklist and complied with the instructions. Grammatical errors in the manuscript has also been corrected. All contributors have been acknowledged including the source of funding for the study